Author’s response to reviews

Title: Percutaneous therapy of a mediastinal lymphangioma with fibrin glue: case report with clinical success after 4 years

Authors:

Siyu Zhou (zgydmay@163.com)
Siyuan Dong (dsycmu@163.com)
Jiang Du (dujiangcmu@163.com)

Version: 3 Date: 16 Jan 2018

Author’s response to reviews:

Dear editor: Thank you for your letter and for the editors’ and reviewer’s comments concerning our manuscript entitled “Percutaneous therapy of a mediastinal lymphangioma with fibrin glue: case report with clinical success after 4 years” (ID: BSUR-D-17-00329R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our research. We have studied comments carefully and have made correction which we hope meet with approval. The main corrections in the paper and the responds to the editors’ comments are as flowing.

1 Consent for publication
It is stated that ‘The manuscript is approved by all authors for publication’. However, it is not a valid Consent for Publication statement. For all manuscripts that include details relating to an individual person, WRITTEN informed consent for the publication of these details must be obtained from that person (or their parent or legal guardian in the case of children under 18). Please provide the Consent for Publication as the following format if you have obtained the Consent for Publication in written form.

"Written informed consent was obtained from the participants for publication of this Case Report and any accompanying tables/images. A copy of the written consent is available for review by the Editor of this journal." Please clarify whether the written informed consent for publication for your manuscript was obtained from the patient. If no, please use the BioMed Central consent form (http://resource-cms.springer.com/springer-cms/rest/v1/content/6621850/data/v1/Consent-Form-PDF) to obtain consent for publication, or a consent form from their own institution or region if appropriate.

Thanks for your valuable suggestions. And we have added this section into our manuscript. The readers may be confused without this part. And the written informed consent for publication for your manuscript was obtained from the patient. Because I am not in my city now, if you need I can send the written informed consent for publication several days latter.

2 Funding
In Funding, please state whether or not the funding body played any roles in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. We revised this part according to your suggestion.

3 Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. We revised this part according to your suggestion.

4 Please name the figure as Figure 1 and refer to each part of the figure as Figure 1a, Figure 1b, etc. in the main text. We revised this part according to your suggestion.